# **July 29**

•	_	•	e.
>T	а	TI	"

Steve, Matt, and Gary.

#### Weather:

Hot and mostly clear, with a few fair weather cumulus and gusty winds in the afternoon.

#### Soundings:

Students launched at 10am and 3pm local. No problems.

### **Checklist:**

Matt checked:

- water level in lidar is good
- web plots
- ✓ real-time lidar displays
- √ 449 power: 2.79 A in the afternoon.

## Cleaning:

Lidar lenses last cleaned by Bill, next cleaning planned Tuesday, Aug 1.

NR01: next cleaning planned Tuesday, Aug 1

#### Other notes:

See previous blog post for note about 449 profiler being briefly turned off this morning. Webplots are now working, the news of which Bill shared with the PI's yesterday. Otherwise all seems to be well.